Author's response to reviews

Title: Met-Independent Hepatocyte Growth Factor-mediated regulation of cell adhesion in human prostate cancer cells

Authors:

Amanda Tate (awtate@emory.edu)
Shuji Isotani (sisotan@med.teikyo-u.ac.jp)
Michael J Bradley (MjB@aol.com)
Robert A Sikes (rasikes@mail.udel.edu)
Rodney Davis (rdavis1@tulane.edu)
Leland WK Chung (lwchung@emory.edu)
Magnus Edlund (edlund@emory.edu)

Version: 3 Date: 24 July 2006

Author's response to reviews:

We have now made our final changes throughout the manuscript. The changes are listed below:

Running title has been remove from the title page.
Results and Discussion headings in the abstract has been renamed
Keywords section has been removed
Section headings has been changed according to instruction.
No underlined text through out the manuscript.

In addition we have made multiple minor changes in punctuation, spacing and text grammar.

Sincerely,
Magnus Edlund